Author’s response to reviews

Title: Delayed Diagnosis of Acromegaly in a Patient with Focal Segmental Glomerulosclerosis: A Rare Case Report and Literature Review

Authors:

Jia Zheng (zhengjiapumc@163.com)
Zhao Cui (cuizhao@bjmu.edu.cn)
Jicheng Lv (jichenglv75@gmail.com)
Hong-zhou Duan (duanhongzhou@126.com)
Suxia Wang (suxiawang@bjmu.edu.cn)
Jun-qing Zhang (junqing.zhang@pkufh.com)
Fu-de Zhou (zhoufude1801@vip.sina.com)
Xiao-hui Guo (bdyyguoxiaohui@sina.com)
Ming-hui Zhao (mhzhao@bjmu.edu.cn)

Version: 1 Date: 03 Jun 2019

Author’s response to reviews:

We have processed the similarity report carefully and made corrections that we hope meet with your approval. To avoid misconduct, we have deleted any unnecessary similar sections and re-write "point-by-point" the similar sections and sentences. Any remaining similar papers were correctly cited.